Author's response to reviews

**Title:** Clinical and physiological effects of transcranial electrical stimulation position on motor evoked potentials in scoliosis surgery

**Authors:**

- YL Lo (gnrlyl@sgh.com.sg)
- YF Dan (gnrdyf@sgh.com.sg)
- YE Tan (gnrlab@sgh.com.sg)
- A Teo (gnrat@sgh.com.sg)
- SB Tan (gostsb@sgh.com.sg)
- WM Yue (gowmy@sgh.com.sg)
- CM Guo (gosgcm@sgh.com.sg)
- S Fook-Chong (gcrfcs@sgh.com.sg)

**Version:** 4  **Date:** 10 February 2010

**Author's response to reviews:** see over
10 February 2010

Dr Theodore Grivas
Editor
Scoliosis Journal

Dear Dr Grivas,

Re: MS: 1184802530319667

Thank you again for all the helpful comments and suggestions.

As advised, I have made the necessary declarations, format changes, reference style adjustments and removal of underlines.
A revised version has been uploaded with this letter. Please do not hesitate to contact me for further clarifications.

Yours sincerely,

YL Lo